Author's response to reviews

Title: Multifocal invasive ductal breast cancer with osteoclast-like giant cells: a case report

Authors:

Georg Richter (richter@pathologie-hannover.de)
Christoph Uleer (christoph.uleer@t-online.de)
Thomas Noesselt (noesselt@kreiskrankenhaus-hameln.de)

Version: 9 Date: 19 November 2010

Author's response to reviews:

The Abstract now matches the Abstract in the submission system.
The patients ethnicity is cited in the case presentation section of the manuscript.
All references are now cited within the main text.
Acknowledgements have been included.
The country has been included in the affiliation.
The manuscript has been checked for typographical errors and final corrections and revisions have taken place.